Author's response to reviews

Title: Protocol for a randomised controlled trial of risk screening and early intervention comparing child- and family-focused cognitive-behavioural therapy for PTSD in children following accidental injury.

Authors:

Justin Kenardy (j.kenardy@uq.edu.au)
Vanessa Cobham (vanessa@psy.uq.edu.au)
Reginald DV Nixon (reg.nixon@flinders.edu.au)
Brett McDermott (Brett.McDermott@mater.org.au)
Sonja March (sonja.march@uq.edu.au)

Version: 2 Date: 30 October 2010

Author's response to reviews:

To whom it may concern,

Thank you for your recent feedback regarding our paper entitled “Protocol for a randomised controlled trial of risk screening and early intervention comparing child- and family-focused cognitive-behavioural therapy for PTSD in children following accidental injury.”

We have made the changes requested and are resubmitting the final version of the manuscript for publication. Below, we outline the changes made to this document.

1. Affiliations. Each author affiliation has been amended on the title page to follow the requested format.

2. All authors have read and approved the final manuscript and we have stated so in the “Author Contributions” section.

3. We have checked the manuscript for typographical errors and made final minor corrections to the manuscript.

Please don’t hesitate to contact us if you require any further changes or information.

We thank you for reviewing and accepting our study protocol paper.

Regards,

Sonja March
(on behalf of the Research Team)